Author’s response to reviews

Title: Design and evaluation of a mobile application to assist the self-monitoring of the chronic kidney disease in developing countries

Authors:

Alvaro Sobrinho (alvaro.sobrinho@ufersa.edu.br)
Leandro Dias da Silva (leandrodias@ic.ufal.br)
Angelo Perkusich (perkusic@dee.ufcg.edu.br)
Maria Eliete Pinheiro (elietepinheiro@uol.com.br)
Paulo Cunha (paulo.cunha65@gmail.com)

Version: 2 Date: 17 Dec 2017

Author’s response to reviews:

The reviewers accepted the article and did not ask for modifications.